\def\papertitle{LOWER AND UPPER SOLUTIONS FOR CONFORMABLE FRACTIONAL DIFFERENTIAL EQUATIONS }
\def\paperauthorA{Bouharket Bendouma}
%\def\paperauthorB{Author_B_fullname}
% you can add more authors using alphabets after \paperauthor
%
%------------------------------------------------------------------------------------------
% Do not delete this part starting from here and up to .....
\documentclass[oneside]{article}
\usepackage{icma2021,amssymb,amsmath,mathrsfs}
%\setcounter{page}{1}
%\input{../../expages.tex}\setpagenumber{1}
%\input{expages.tex}\setpagenumber{1}
\ninept
%\usepackage[notref,notcite]{showkeys}
\usepackage{mathptmx}
% Saves a lot of ouptut space in PDF... after conversion with the distiller
% Delete if you cannot get PS fonts working on your system.

% pdf-tex settings: detect automatically if run by latex or pdflatex
\newif\ifpdf
\ifx\pdfoutput\relax \else \ifcase\pdfoutput \pdffalse \else
\pdftrue \fi

\ifpdf % compiling with pdflatex
\usepackage[pdftex,
pdftitle={\papertitle},
%pdfauthor={\paperauthorA},
% for more than one author use pdfauthor={\paperauthorA,\paperauthorB, .....},
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%ù
colorlinks=false, % links are activated as colror boxes instead of color text
bookmarksnumbered, %use section numbers with bookmarks
pdfstartview= XYZ   % starts with zoom=100% instead of full screen; especially useful if working with a big screen :-)
]{hyperref} \pdfcompresslevel=9
\usepackage[pdftex]{graphicx}
\usepackage[figure,table]{hypcap}
\else % compiling with latex
\usepackage[dvips]{graphicx}
\usepackage[dvips,
colorlinks=false,           % no color links
bookmarksnumbered, %use section numbers with bookmarks
pdfstartview= XYZ   % starts with zoom=100% instead of full screen
]{hyperref}
% hyperrefs are active in the pdf file after conversion
\usepackage[figure,table]{hypcap}
\fi

\newcommand\dsum{\displaystyle\sum}
\newcommand\dprod{\displaystyle\prod}
%------------------------------------------------------------------------------------------
\title{\papertitle}
\affiliation{\paperauthorA} % This command replaces \name{The DAFx Crew}
{Faculty of Mathematics and Informatics, Ibn Khaldoun, Tiaret University,\\
PoBox 78, 14000 za\^{a}roura, Tiaret, Algeria}

%\usepackage{layout}
%------------------------------------------------------------------------------------------
%%%%%%%%%%%%%%%%%%%%%%% to here %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{99}

\bibitem{IyNw}
O.S. Iyiola \ and\ E.R.Nwaeze, Some new results on the new
conformable fractional calculuswith application using D�Alambert
approach, Progr. Fract. Di�er. Appl.,(2016), 2(2), 115--122.

\bibitem{KhHoSa}
R. Khalil, M. Al Horani, A. Yousef, M. Sababheh,\emph{ A new
definition of fractional derivative}, J. Comput. Appl. Math. {\bf
264} (2014), 65--70.

\bibitem{KiSrTr} A. Kilbas, M.H. Srivastava and J.J. Trujillo,\emph{ Theory and application of fractional differential equations}, North
Holland Mathematics Studies 204, 2006.

%\bibitem{Ortig}M.D. Ortigueira,\emph{ Fractional Calculus for Scientists and Engineers}, Lecture Notes in Electrical Engineering, 84. Springer, Dordrecht, 2011.
\bibitem{Nwa}E.R. Nwaeze, \emph{A Mean Value Theorem for the Conformable Fractional Calculus
on Arbitrary Time Scales}, j. Progr. Fract. Differ. Appl. {\bf
2}(2016), no. 4, 287--291.

\bibitem{Podl}
I. Podlubny,\emph{ Fractional differential equations}, Academic
Press: San Diego CA, (1999).

%\bibitem{RaVa} J.D. Ramirez and A.S. Vatsala, \emph{Monotone iterative technique for fractional differential equations with periodic boundary conditions}, Opuscula Mathematica, vol. 29, no. 3, 2009, 289�-304,

\bibitem{SaKiMa}
S.G. Samko, A.A. Kilbas\ and\ O.I. Marichev, \emph{Fractional
integrals and derivatives: Theory and applications}, Gordon and
Breach, Yverdon, 1993.

\bibitem{ShHuYaYi} K. Shugui, C. Huiqing, Y. Yaqing and G. Ying,\emph{ Existence and uniqueness of the solutions for the fractional
initial value problem},{ Electronic Journal of Shanghai Normal
University (Natural Sciences)}, vol. 45, no.3, 2016, 313-319, .

\end{thebibliography}


    %\layout

    % more pdf-tex settings:
    \ifpdf % used graphic file format for pdflatex
    \DeclareGraphicsExtensions{.png,.jpg,.pdf}
    \else  % used graphic file format for latex
    \DeclareGraphicsExtensions{.eps}
    \fi

    \maketitle

\begin{abstract}
In this article, we study the existence of solutions to conformable
fractional differential equations with periodic boundary condition.
Existence results are obtained by the method of upper and lower
solutions and Schauder's fixed point theorem.
\end{abstract}

\section{Introduction}
Fractional differential equations plays an important role in
describing many phenomena and processes in various fields of science
such as physics, chemistry, control systems, population dynamics,
etc., see \cite{KiSrTr, Podl, SaKiMa}. Recently, a new fractional
derivative called the conformable fractional derivative was
introduced by Khalil et al. \cite{KhHoSa}.\vspace{0.1cm}

\ In this paper, we are concerned with the existence of a solutions
for the following conformable fractional differential equations with
periodic boundary condition:
\begin{equation}\label{eq1.1}
   \begin{cases}
      x^{(\alpha)}(t)=f(t,x(t)),\quad \ \ \ for\ t\in J=[a,b],\ a>0,\vspace{0.2cm}\\
      x(a)=x(b).
   \end{cases}
\end{equation}
Where $0<\alpha\leq1$, $f:J\times \mathbb{R}\rightarrow \mathbb{R}$
is a continuous function, $x^{(\alpha)}(t)$ denotes the conformable
fractional derivative of $x$ at $t$ of order $\alpha$.\vspace{0.1cm}

\ The paper is organized as follows. In Section~\ref{sec:2}, we
present the main concepts of conformable fractional derivative
calculus and we give some useful preliminary results. In
Section~\ref{Sec:3}, we prove existence of solutions to problem
\eqref{eq1.1} by using the method of upper and lower solutions and
Schauder's fixed point theorem.

\section{Preliminaries}
\label{sec:2}

In this section, we introduce the definition of conformable
fractional calculus and their important properties.
\begin{definition}\cite{KhHoSa}
\label{d:cfd} Given a function $f : [0,\infty)\rightarrow
\mathbb{R}$ and a real constant $\alpha\in(0,1]$. The conformable
fractional derivative of $f$ of order $\alpha$ is defined by,
\begin{equation}f^{(\alpha)}(t):=\lim_{\varepsilon\rightarrow0} \frac{f(t+\varepsilon
t^{1-\alpha})-f(t)}{\varepsilon}
\end{equation}
for all $t>0$.\ If $f^{(\alpha)}(t)$ exists and is finite, we say
that $f$ is $\alpha$-differentiable at $t$.

If $f$ is $\alpha$-differentiable in some interval $(0,a), \ a>0$,
and $\lim_{t\rightarrow 0^+}f^{(\alpha)}(t)$ exists, then the
conformable fractional derivative of $f$ of order $\alpha$ at $t=0$
is defined as
$$f^{(\alpha)}(0)=\lim_{t\rightarrow0^+}f^{(\alpha)}(t).$$
\end{definition}

\begin{theorem}\cite{KhHoSa}\label{th:00}
Let $\alpha\in(0,1]$ and $f:[0,\infty)\rightarrow \mathbb{R}$ a
$\alpha$-differentiable function at $t_{0} > 0$,  then $f$ is
continuous at $t_{0}$.
\end{theorem}

\begin{theorem}\cite{KhHoSa}
\label{th:01} Let $\alpha\in(0,1]$ and assume $f,g$ to be
$\alpha$-differentiable at a point $t > 0$. Then,
\begin{enumerate}
\item[(i)] $(a f+b g)^{(\alpha)}=af^{(\alpha)}+b g^{(\alpha)}$, \ \ for all $a,b\in\mathbb{R}$;
\item[(ii)] $(f g)^{(\alpha)}=f g^{(\alpha)}+g f^{(\alpha)}$;
\item[(iii)] $\left(f/g\right)^{(\alpha)}=\dfrac{g f^{(\alpha)}-f g^{(\alpha)}}{g^{2}}$.
\item[(iv)] If, in addition, $f$ is differentiable at a point $t>0$, then
$$f^{(\alpha)}(t)=t^{1-\alpha}f'(t).$$
\end{enumerate}
\end{theorem}
 Additionaly, conformable fractional derivatives of certain
 functions as follow:
\begin{enumerate}
\item $(t^{p})^{(\alpha)}=p\,t^{p-\alpha}$,  for all $p\in\mathbb{R}$.
\item $(\lambda)^{(\alpha)}=0$, for all  $\lambda \in \mathbb{R}$.
\item $(e^{ct})^{(\alpha)}=c\,t^{1-\alpha}e^{ct}$,   for all $c\in\mathbb{R}$.
\item $(e^{\frac{p}{\alpha}t^{\alpha}})^{(\alpha)}=p\,e^{\frac{p}{\alpha}t^{\alpha}}$,  for all $p\in\mathbb{R}$.
\end{enumerate}

\begin{remark}
If $f$ is differentiable at $t$, then f is $\alpha$-differentiable
at $t$.
\end{remark}
We introduce the following space:
\begin{eqnarray*}
C^{\alpha}(J,\mathbb{R})&=& \{f:J\rightarrow \mathbb{R}, \ \mbox{ is
\ $\alpha$-differentiable on $J$ \ and} \; f^{(\alpha)}\in
C(J,\mathbb{R})\}.
\end{eqnarray*}

\begin{definition}$($Conformable fractional integral \cite{KhHoSa}$)$.\label{def:cfI}
Let $\alpha\in(0,1]$ and $f: [a,\infty)\rightarrow \mathbb{R}$. The
conformable fractional integral of $f$ of order $\alpha$ from $a$ to
$t$, denoted by $I_{\alpha}^{a}(f)(t)$, is defined by
$$
I_{\alpha}^{a}(f)(t)
:=\int_{a}^{t}f(s)d_{\alpha}s:=\int_{a}^{t}f(s)s^{\alpha-1}ds.
$$
The considered integral is the usual improper Riemann one.
\end{definition}

\begin{theorem}\cite{KhHoSa}
\label{th:02} If $f$ is a continuous function in the domain of
$I_{\alpha}^{a}$ then, for all $t\geq a$ we have
$$\left(I_{\alpha}^{a}(f)\right)^{(\alpha)}(t)=f(t).$$
\end{theorem}


\begin{lemma}\cite{KhHoSa} \label{propo}
Let $f:(a,b) \rightarrow \mathbb{R}$ be differentiable and $0<
\alpha \leq 1$. Then, for all $t>a$ we have
\begin {equation}
I_\alpha^{a} (f^{(\alpha)})(t)= f(t)-f(a).
  \end{equation}
\end{lemma}
\begin{proposition}\cite{IyNw}
\label{pr1}Let $0 < a < b$, $f:[a, b]\rightarrow \mathbb{R}$ be
continuous function and $0< \alpha < 1$. Then for all $t\in[a, b]$
we have,
\[\left|I_{\alpha}^{a}(f)(t)\right| \leq I_{\alpha}^{a}|f|(t).\]
\end{proposition}

\section{Main Results}
\label{Sec:3} In this section, we establish an existence result for
the problem (\ref{eq1.1}). A solution of problem (\ref{eq1.1}) will
be a function $ x\in C^{\alpha}(J,\mathbb{R})$ for which
(\ref{eq1.1}) is satisfied. We introduce the notion of lower and
upper solutions for the problem \eqref{eq1.1}.
\begin{definition} \label{def2.1} \rm
Let $v \in C^{\alpha}(J,\mathbb{R})$. We say that $v $ is a lower
solution of the boundary value problem \eqref{eq1.1}, if
\begin{itemize}
    \item[(i)] $v^{(\alpha)}(t)\leq f(t,v(t)), \quad \mbox{for all } t \in J$;
    \item[(ii)] $v(a)\leq v(b).$
 \end{itemize}
Let $w\in C^{\alpha}(J,\mathbb{R})$. We say that $w$ is an upper
solution of the boundary value problem \eqref{eq1.1}, if
\begin{itemize}
    \item[(i)] $w^{(\alpha)}(t)\geq f(t,w(t)), \quad \mbox{for all } t \in J$;
    \item[(ii)] $w(a)\geq w(b).$
 \end{itemize}
\end{definition}
we define the sector
$$[v,w] =\{x \in C(J,\mathbb{R}): v(t)\leq x(t)\leq w(t), \; \mbox{for all} \;t\in J \}.$$
We consider the following problem.
\begin{equation}\label{E3.1}
\begin{cases}
  x^{(\alpha)}(t)+ \alpha x(t)=f(t,\bar{x}(t))+\alpha \bar{x}(t), \ \ \ t\in  J,\vspace{0.25cm}\\
 x(a)=x(b). \end{cases}
\end{equation}
where
\begin{equation}\label{E3.02}
 \bar{x}(t)=\left\{
\begin{array}{ll}
  v(t),& \ \ \ \ \ \ \  \ \ \ \ \ \textrm{if $x(t)<v(t)$},\vspace{0.25cm} \\
  x(t),& \ \ \ \ \ \ \ \ \ \ \ \ \ \textrm{if $v(t)\leq x(t)\leq w(t)$},\vspace{0.25cm} \\
  w(t),& \ \ \ \ \ \ \ \ \ \ \ \ \ \textrm{if $x(t)>w(t)$},
\end{array}\right.
\end{equation}
and $v(t)\leq w(t)$ on $J$.\\
We need the following auxiliary proposition.

\begin{proposition}\label{p2.08} For every $g\in C^{\alpha}(J,\mathbb{R})$, the
problem
\begin{equation}\label{EP.031}
   \begin{cases}
     x^{(\alpha)}(t)+ \alpha x(t)=g(t), & t\in J=[a,b], \quad a>0,\quad \; 0<\alpha<1,\vspace{0.2cm}\\
     x(a)=x(b).
   \end{cases}
\end{equation}
has a unique solution $x \in C^{\alpha}(J,\mathbb{R})$ given by:
\begin{equation}
\label{eq.th02}
x(t):=e^{-t^{\alpha}}\left(\dfrac{1}{e^{b^{\alpha}-a^{\alpha}}-1}
 \int_{a}^{b}s^{\alpha-1}e^{s^{\alpha}}g(s)ds +
 \int_{a}^{t}s^{\alpha-1}e^{s^{\alpha}}g(s)ds\right)
\end{equation}
\end{proposition}

Let us define the operators $ \mathcal{N}:C(J,\mathbb{R})\rightarrow
C(J,\mathbb{R})$ by
\begin{align*} \mathcal{ N}(x)(t)=e^{-t^{\alpha}}\left(\dfrac{1}{e^{b^{\alpha}-a^{\alpha}}-1}
 \int_{a}^{b}s^{\alpha-1}e^{s^{\alpha}}\Big(f(s,\bar{x}(s))+\alpha \bar{x}(s)\Big)ds \right. \vspace{0.2cm}\\
\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \
\left.+
\int_{a}^{t}s^{\alpha-1}e^{s^{\alpha}}\Big(f(s,\bar{x}(s))+\alpha
\bar{x}(s)\Big)ds \right).\end{align*}

\begin{proposition}\label{P3.002}
If $ v, w \in C^{\alpha}(J,\mathbb{R})$ are, respectively, lower and
upper solutions to \eqref{eq1.1}, such that $v(t)\leq w(t)$, then
the operator $\mathcal{N}$ is compact.
\end{proposition}


Now, we can obtain the main theorem of this section.

\begin{theorem}\label{T3.2}
If $ v, w \in C^{\alpha}(J,\mathbb{R})$ are, respectively, lower and
upper solutions to (\ref{eq1.1}), such that $v(t)\leq w(t)$, then
the problem (\ref{eq1.1}) has a solution $x\in
C^{\alpha}(J,\mathbb{R})$, such that $v(t)\leq x(t)\leq w(t)$ for
all $t\in J$.
\end{theorem}

\begin{example} \label{x2-t} \rm %(3.9)
Consider the periodic problem:
\begin{equation}\label{ES.03}
 \begin{cases}
      x^{(1/3)}(t)=\dfrac{2t-1-x^{5}(t)}{\sqrt{t}}    \  \ \ t \in J=[1/2,1],  \vspace{0.25cm}\\
      x(1/2)= x(1).
  \end{cases}
\end{equation}
Here $n=1$,
  $\alpha=1/3$, and $f(t,x)=\dfrac{2t-1-x^{5}(t)}{\sqrt{t}}$. It is clear that $f$ is a continuous function. Observe that  $v=-1 $ and $w =1$ are,
respectively, lower and upper solutions of (\ref{ES.03}) follows
from the fact that
\[v^{(\frac{1}{3})}(t)=0\leq f(t,v(t))=2\sqrt{t}, \;t \in J,\ \ \ v(1/2)\leq v(1),\]
and
\[w^{(\frac{1}{3})}(t)=0 \geq f(t,w(t))=\frac{2(t-1)}{\sqrt{t}}, \; t\in J, \ \ \ w(1/2)\geq w(1),\]
Theorem \ref{T3.2}, implies that problem \eqref{ES.03} has a
solution $x\in c^{\frac{1}{3}}([1/2,1])$, such that $-1 \le x(t) \le
1$, for all $t \in [1/2 ,1]$.

\end{example}
\section{Conclusion}
In this paper, we present existence of solutions for the conformable
fractional differential equations with periodic boundary condition.
These results are obtained by using the notion of upper and lower
solutions and Schauder's fixed point theorem.



\end{document}
